import { TypeList } from "docs-ui"

# PromotionUtils

## Enumerations

- [ApplicationMethodAllocation](../PromotionUtils/enums/utils.PromotionUtils.ApplicationMethodAllocation/page.mdx)
- [ApplicationMethodTargetType](../PromotionUtils/enums/utils.PromotionUtils.ApplicationMethodTargetType/page.mdx)
- [ApplicationMethodType](../PromotionUtils/enums/utils.PromotionUtils.ApplicationMethodType/page.mdx)
- [CampaignBudgetType](../PromotionUtils/enums/utils.PromotionUtils.CampaignBudgetType/page.mdx)
- [ComputedActions](../PromotionUtils/enums/utils.PromotionUtils.ComputedActions/page.mdx)
- [PromotionActions](../PromotionUtils/enums/utils.PromotionUtils.PromotionActions/page.mdx)
- [PromotionRuleOperator](../PromotionUtils/enums/utils.PromotionUtils.PromotionRuleOperator/page.mdx)
- [PromotionStatus](../PromotionUtils/enums/utils.PromotionUtils.PromotionStatus/page.mdx)
- [PromotionType](../PromotionUtils/enums/utils.PromotionUtils.PromotionType/page.mdx)
- [RuleType](../PromotionUtils/enums/utils.PromotionUtils.RuleType/page.mdx)
